Author's response to reviews

Title: Co-expression of putative stemness and epithelial-to-mesenchymal transition markers on single circulating tumour cells from patients with early and metastatic breast cancer

Authors:

Maria A Papadaki (papadaki_maria1@yahoo.gr)
Galatea Kallergi (kalerqi@med.uoc.gr)
Zafeiris Zafeiriou (Zaphiris_Z@yahoo.de)
Lefteris Manouras (leman.9@hotmail.com)
Panayiotis A Theodoropoulos (takis@med.uoc.gr)
Dimitris Mavroudis (mavrudis@med.uoc.gr)
Vassilis Georgoulias (georgsec@med.uoc.gr)
Sofia Agelaki (agelaki@med.uoc.gr)

Version: 10  Date: 27 August 2014

Author's response to reviews: see over
Dear Dr Maria Grazia Daidone,

Thank you for your email concerning the revision of our manuscript entitled “Co-expression of putative stemness and epithelial-to-mesenchymal transition markers on single circulating tumour cells from patients with early and metastatic breast cancer” by Maria A Papadaki, Galatea Kallergi, Zafeiris Zafeiriou, Lefteris Manouras, Panayiotis A Theodoropoulos, Dimitris Mavroudis, Vassilis Georgoulias, Sofia Agelaki.

According to the editor’s comment, we have now included a sentence in the Methods-Patient samples section (page 7), to verify that healthy volunteers also gave their written informed consent to participate in the study.

I remain at your disposal for any additional information you may request.

Yours Sincerely,

Dr Galatea Kallergi